Author's response to reviews

Title: Translocation pathways for inhaled asbestos fibers

Authors:

Giuseppe A Miserocchi (giuseppe.miserocchi@unimib.it)
Giulio A Sancini (giulio.sancini@unimib.it)
Francesco Mantegazza (francesco.mantegazza@unimib.it)
Gerolamo Chiappino (fibre@unimi.it)

Version: 4 Date: 19 December 2007

Author's response to reviews:

We thank the editorial assistant for review. We have made all the corrections suggested, shortened the Abstract (it is now 303 words for a maximum allowed of 350 words) and rewritten some parts to improve the english. We would like to retain the paragraphs headings because we think that would help to follow the presentation. Hope the present version will be satisfactory.